Author’s response to reviews

Title: Clinical relevance of contextual factors as triggers of placebo and nocebo effects in musculoskeletal pain

Authors:

Giacomo Rossettini (giacomo.rossettini@gmail.com)

Elisa Carlino (elisa.carlino@gmail.com)

Marco Testa (marco.testa@unige.it)

Version: 2 Date: 14 Jan 2018

Author’s response to reviews:

1. As per our submission guidelines for Debate articles, the abstract should not contain separate section, therefore please remove the "Background", "Discussion" and "Summary" subheadings

Thanks. We have modified the “Abstract” as you suggested.

2. Please remove the table of contents as this is not required in your manuscript.

Thanks. We have removed the “Table of contents” as you requested.

3. Please clarify whether the figures (1-4) used in your manuscript are your own figures or obtained from another source. Please note that if these figures are obtained from an external source, a reference should be provided.

The images are our property and they are original.

4. Please remove the response to reviewers document as this is not required at this stage.

Ok.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Ok.